Author’s response to reviews

Title: Electronic voting to encourage interactive lectures: a randomised trial

Authors:

Paul M Duggan (paul.duggan@adelaide.edu.au)
Edward Palmer (edward.palmer@adelaide.edu.au)
Peter Devitt (peter.devitt@adelaide.edu.au)

Version: 4 Date: 22 July 2007

Author’s response to reviews:

This final revision covers requested formatting changes only